Author's response to reviews

**Title:** Test-Retest Variability of High Resolution Positron Emission Tomography (PET) Imaging of Cortical Serotonin (5HT2A) Receptors in Older, Healthy Adults

**Authors:**

Tiffany W Chow (tchow@rotman-baycrest.on.ca)
David C Mamo (David_Mamo@camh.net)
Hiroyuki Uchida (Hiroyuki_Uchida@camh.net)
Sylvain Houle (sylvain.houle@camhpet.ca)
Gwenn S Smith (Gwenn_Smith@camh.net)
Bruce G Pollock (Bruce_Pollock@camh.net)
Benoit H Mulsant (Benoit_Mulsant@camh.net)

**Version:** 2  **Date:** 22 August 2008

**Author's response to reviews:**

Although your online system states that we are not eligible for waiving the publication fees, I believe the University of Toronto, where all co-authors are based, qualifies. Pls see the supplementary attachment, which is an email from BMC that addresses this issue in 2006 for another submission.